

## 

**Citation:** Cama A, Müller A, Taoaba R, Butcher RMR, Itibita I, Migchelsen SJ, et al. (2017) Prevalence of signs of trachoma, ocular *Chlamydia trachomatis* infection and antibodies to Pgp3 in residents of Kiritimati Island, Kiribati. PLoS Negl Trop Dis 11(9): e0005863. https://doi.org/10.1371/ journal.pntd.0005863

Editor: Michael French, RTI International, UNITED STATES

Received: March 8, 2017

Accepted: August 8, 2017

Published: September 12, 2017

**Copyright:** © 2017 World Health Organization. Licensee Public Library of Science. This is an open access article distributed under the Creative Commons Attribution IGO License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. http://creativecommons.org/licenses/by/3.0/ igo/. In any use of this article, there should be no suggestion that WHO endorses any specific organization, products or services. The use of the WHO logo is not permitted. This notice should be preserved along with the article's original URL.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

RESEARCH ARTICLE

# Prevalence of signs of trachoma, ocular *Chlamydia trachomatis* infection and antibodies to Pgp3 in residents of Kiritimati Island, Kiribati

Anaseini Cama<sup>1,2</sup>, Andreas Müller<sup>3</sup>, Raebwebwe Taoaba<sup>4</sup>, Robert M. R. Butcher<sup>5</sup>\*, lakoba Itibita<sup>6</sup>, Stephanie J. Migchelsen<sup>5</sup>, Tokoriri Kiauea<sup>6</sup>, Harry Pickering<sup>5</sup>, Rebecca Willis<sup>7</sup>, Chrissy h. Roberts<sup>5</sup>, Ana Bakhtiari<sup>7</sup>, Richard T. Le Mesurier<sup>1,2</sup>, Neal D. E. Alexander<sup>8</sup>, Diana L. Martin<sup>9</sup>, Rabebe Tekeraoi<sup>4</sup>, Anthony W. Solomon<sup>5,10</sup>, for the Global Trachoma Mapping Project<sup>1</sup>

 International Agency for the Prevention of Blindness, Western Pacific Region, Suva, Fiji, 2 The Fred Hollows Foundation, Sydney, Australia, 3 Centre for Eye Research, University of Melbourne, Melbourne, Australia, 4 Eye Department, Ministry of Health and Medical Services, South Tarawa, Kiribati, 5 Clinical Research Department, London School of Hygiene & Tropical Medicine, London, United Kingdom, 6 Kiritimati Hospital, London, Kiritimati Island, Kiribati, 7 International Trachoma Initiative, Task Force for Global Health, Decatur, Georgia, United States of America, 8 MRC Tropical Epidemiology Group, Infectious Disease Epidemiology Department, London School of Hygiene & Tropical Medicine, London, United Kingdom,
 9 Division of Parasitic Diseases and Malaria, Centers for Disease Control and Prevention, Atlanta, Georgia, United States of America, 10 Department of Control of Neglected Tropical Diseases, World Health Organization, Geneva, Switzerland

¶ Membership of The Global Trachoma Mapping Project is provided in the Acknowledgments. \* robert.butcher@lshtm.ac.uk

# Abstract

### Objective

In some Pacific Island countries, such as Solomon Islands and Fiji, active trachoma is common, but ocular *Chlamydia trachomatis* (*Ct*) infection and trachomatous trichiasis (TT) are rare. On Tarawa, the most populous Kiribati island, both the active trachoma sign "trachomatous inflammation—follicular" (TF) and TT are present at prevalences warranting intervention. We sought to estimate prevalences of TF, TT, ocular *Ct* infection, and anti-*Ct* antibodies on Kiritimati Island, Kiribati, to assess local relationships between these parameters, and to help determine the need for interventions against trachoma on Kiribati islands other than Tarawa.

### Methods

As part of the Global Trachoma Mapping Project (GTMP), on Kiritimati, we examined 406 children aged 1–9 years for active trachoma. We collected conjunctival swabs (for droplet digital PCR against *Ct* plasmid targets) from 1–9-year-olds with active trachoma, and a systematic selection of 1–9-year-olds without active trachoma. We collected dried blood spots (for anti-Pgp3 ELISA) from all 1–9-year-old children. We also examined 416 adults aged  $\geq$ 15 years for TT. Prevalence of TF and TT was adjusted for age (TF) or age and gender (TT) in five-year age bands.

Funding: This study was principally funded by the Global Trachoma Mapping Project (GTMP) grant from the United Kingdom's Department for International Development (ARIES: 203145) to Sightsavers, which led a consortium of nongovernmental organizations and academic institutions to support health ministries to complete baseline trachoma mapping worldwide; the Wellcome Trust (098521); and the Fred Hollows Foundation (ref 1954-0). The GTMP was also funded by the United States Agency for International Development, through the ENVISION project implemented by RTI International under cooperative agreement number AID-OAA-A-11-00048, and the END in Asia project implemented by FHI360 under cooperative agreement number OAA-A-10-00051. A committee established in March 2012 to examine issues surrounding completion of global trachoma mapping was initially funded by a grant from Pfizer to the International Trachoma Initiative. AWS was a Wellcome Trust Intermediate Clinical Fellow at the London School of Hygiene & Tropical Medicine, and is now a staff member of the World Health Organization. The authors alone are responsible for the views expressed in this article and they do not necessarily represent the views, decisions or policies of the institutions with which they are affiliated. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** AC and RTLM of the Fred Hollows Foundation were personally involved in the study, as indicated in the statement of authors' contributions; otherwise, funders had no role in project design, in project implementation or analysis or interpretation of data, in the decisions on where, how or when to publish in the peer reviewed press, or in preparation of the manuscript.

#### Results

The age-adjusted prevalence of TF in 1–9-year-olds was 28% (95% confidence interval [CI]: 24–35). The age- and gender-adjusted prevalence of TT in those aged  $\geq$ 15 years was 0.2% (95% CI: 0.1–0.3%). Twenty-six (13.5%) of 193 swabs from children without active trachoma, and 58 (49.2%) of 118 swabs from children with active trachoma were positive for *Ct* DNA. Two hundred and ten (53%) of 397 children had anti-Pgp3 antibodies. Both infection (p<0.0001) and seropositivity (p<0.0001) were strongly associated with active trachoma. In 1–9-year-olds, the prevalence of anti-Pgp3 antibodies rose steeply with age.

#### Conclusion

Trachoma presents a public health problem on Kiritimati, where the high prevalence of ocular *Ct* infection and rapid increase in seropositivity with age suggest intense *Ct* transmission amongst young children. Interventions are required here to prevent future blindness.

### Author summary

Ocular infection with *Chlamydia trachomatis* causes trachoma. It is the leading infectious cause of blindness, and the target of an international campaign for elimination as a public health problem. Trachoma is endemic to Tarawa, the most populated island of Kiribati, housing approximately half the national population. However, the country has 20 inhabited islands and there were no previous trachoma prevalence data from Kiribati outside Tarawa. We set out to determine the prevalence of trachoma in the second most populated island, Kiritimati, located over 3000 km from Tarawa. In some other Pacific Island countries, ocular *C. trachomatis* infection is much less prevalent than the clinical signs that are used to guide interventions; we therefore looked for PCR-based evidence of current infection and antibodies to chlamydial proteins, in addition to recording clinical signs of trachoma. Our results indicate that trachoma and ocular *C. trachomatis* infection are prevalent on Kiritimati, and suggest that interventions are required here. The combined application of antibody, nucleic acid and clinical tools in an intervention-naïve population provides insight into their inter-relationships and the data are, therefore, of considerable interest to elimination programmes within and beyond the Pacific.

### Introduction

Trachoma is the leading infectious cause of blindness and the subject of an international campaign for elimination as a public health problem [1,2]. The strategy for elimination uses the acronym "SAFE": <u>surgery</u> for advanced disease; and <u>antibiotics</u>, <u>facial cleanliness</u> and <u>environ-</u> mental improvement to clear infection and reduce transmission of *Chlamydia trachomatis* (*Ct*), the causative organism.

Australia [3] and several Pacific Island nations [4–6] have long been known to be trachoma-endemic. The first systematic investigation of trachoma in the Pacific was conducted in 2007, when a series of Trachoma Rapid Assessments was undertaken involving examination of 3102 children aged 1–9 years and 903 adults aged  $\geq$ 40 years in 67 high-risk communities of Kiribati, Nauru, Vanuatu, Solomon Islands and Fiji [7]. In Kiribati, the 2007 Trachoma Rapid Assessments included 15 sites in Betio and Buota on Tarawa Atoll, in the far west of the country. A total of 655 children aged 1–9 years and 160 adults aged  $\geq$ 40 years were examined; more than one-third of those children had trachomatous inflammation—follicular (TF), and nearly two-thirds of the adults had trachomatous conjunctival scarring (TS) [8]. One case of trachomatous trichiasis (TT) was identified, and it was reported that 10 TT surgeries had been performed locally (in a population of about 100,000 people) [9] over a twelve-month period. Trachoma Rapid Assessment data are deliberately biased to increase the likelihood that trachoma will be identified where it is endemic and therefore do not provide the prevalence estimates necessary to facilitate programme planning; these results provided justification for undertaking a formal prevalence survey.

In 2012, therefore, a population-based prevalence survey was undertaken in Kiribati's capital, South Tarawa, and the adjacent township of Betio (Fig 1). It estimated the unadjusted TF prevalence in children aged 1-9 years to be 21% and the unadjusted TT prevalence in adults aged >15 years to be 1.5% [10]. Because, in 2010, the combined population of South Tarawa and Betio was 49% of the national population [9], this survey could have been taken as evidence justifying implementation of the A, F and E components of the SAFE strategy [11] throughout Kiribati: even if there was no TF elsewhere in the country, the national TF prevalence would still have been greater than the 10% threshold above which WHO recommends A, F and E for at least three years before a repeat survey [12]. However, Kiribati is made up of 33 atolls and islands dispersed over 3.5 million square kilometers of ocean-approximately half the size of Australia-and the expense of rolling out programme-level interventions in a nation of many tiny islands dispersed over such an area would be considerable. In addition, recent evidence suggests that the epidemiology of trachoma in Vanuatu, Fiji and the Solomon Islands is different to that in much of trachoma-endemic Africa, with a relative paucity of TT [13] and low prevalences of ocular Ct infection for the observed prevalences of TF [14-17]. Whether or not TT and ocular Ct infection were present elsewhere in Kiribati was therefore a question of both public health and scientific significance.

Situated in the far east of Kiribati, Kiritimati Island of the northern Line Islands (Fig 1) has the greatest land area (388 km<sup>2</sup>) of any coral atoll in the world and comprises over 70% of the country's total land area. It is the second-most highly populated island of Kiribati after Tarawa [9], with an estimated population in 2010 of 5586 people. We sought to estimate the prevalences of trachoma, ocular *Ct* infection, and anti-*Ct* antibodies on Kiritimati Island, Kiribati, in part to assess the relationships between these indices locally, and in part to guide planning processes for a national trachoma elimination programme.

#### **Methods**

#### **Ethics statement**

Ethics approval was received from the Research Ethics Committee of the London School of Hygiene & Tropical Medicine (reference numbers 6319, 8355 and 10136) and the Kiribati Ministry of Health and Medical Services (08/11/2015). Verbal consent to involve each village was obtained from its leaders. Written informed consent was obtained from all individuals examined, or a parent or guardian if the participant was aged <15 years.

#### Study design

Children aged 1–9 years are the standard indicator group for estimating the prevalence of active trachoma [12]. We wished to have 95% confidence of estimating an expected TF prevalence in 1–9-year-olds of 20% with absolute precision of  $\pm$ 5%. Using the single population proportion for precision formula and correcting for a small, finite population (because the



**Fig 1.** Map of Kiribati, with zoomed maps of Tarawa (lower left panel) and Kiritimati (lower right panel). Produced in Miller projection using QGIS 2.16. Shapefiles from gadm.org.

https://doi.org/10.1371/journal.pntd.0005863.g001

estimated number of 1–9-year-olds resident on Kiritimati was 1222), and using a design effect of 2.65, our sample size estimate was 426 1–9-year-olds [9,18,19]. Although we determined the sample size as a number of children, all individuals aged  $\geq$ 1 year resident in selected clusters and who gave consent to be examined were included in the survey [19]. All four inhabited villages (the smallest administrative unit) on Kiritimati Island were visited (Fig 1), with the number of households to be invited to participate in each village proportional to that village's relative size: 62 households from London, 106 households from Tabwekea, 42 households from Banana, and 11 households from Poland (Fig 1). Maps of each village were prepared by hand and compact segment sampling used, with clusters of households selected by random draw. In addition to household visits, data were collected at Presbyterian and Catholic *maneabas* (large open community halls where families gather and often stay for months) within or adjacent to the selected clusters.

#### **Clinical examination**

Graders were trained and certified according to the standardized training protocols of the GTMP [19]. Trachoma grading was undertaken according to the WHO simplified grading

scheme [8], using  $2.5 \times$  binocular magnifying loupes and sunlight illumination. Eyes with trichiasis were considered to have TT if and only if they also had TS. (The presence or absence of TS was not recorded in the absence of trichiasis.) Graders cleaned their hands with alcoholbased hand gel after each examination.

Participants identified as having active trachoma (TF and/or trachomatous inflammation intense [TI] in one or both eyes) were provided with two tubes of 1% tetracycline eye ointment, and they or their parents or guardian were instructed on how to apply it. Participants with trichiasis were referred for management by a trained ophthalmologist.

### Specimen collection and handling

When a 1–9-year-old child was found to have TF and/or TI, a conjunctival swab was taken by the examiner. In addition, a conjunctival swab was taken from a systematic two-thirds sample of 1–9-year-old children without these signs, with the creation of this sub-sampling approach necessitated by the discovery that a box of sterile swabs had been misplaced in transit to the island, prior to fieldwork-this meant that we would not be able to swab all the children that we intended to examine. Systematic sampling was achieved by selecting the first two of every three children without active trachoma, in order of presentation. Individuals older than 9 years were not swabbed, regardless of whether or not they had active trachoma. In subjects selected for swabbing, a single sterile polyester-coated swab (Puritan Medical Products, Guilford, USA) was passed four times across the upper tarsal conjunctiva of the more inflamed eye, rotating the swab 90° after each pass. Using a technique that minimised the likelihood of the retained part of the swab shaft or swab-head touching anything other than the subject's conjunctiva and the collection tube, swabs were placed in polypropylene tubes and refrigerated at the end of each day; ice was not available locally. Following grading and (if indicated) swab collection, examiners collected fingerpick blood onto filter paper (Cell Labs Pty, Sydney, Australia) from all 1-9-year-old children assessed. Filter papers were air-dried overnight in an airconditioned room, then packed in individual resealable plastic bags; up to 100 of those primary plastic bags were enclosed in large secondary resealable plastic bags containing dessicant sachets for refrigeration and transport. Swabs and dried blood spots were shipped at ambient temperature to the London School of Hygiene & Tropical Medicine for droplet digital PCR (ddPCR) and anti-Ct antibody testing, respectively. Handling and processing was identical for all samples, being undertaken masked to examination results.

### ddPCR for ocular Chlamydia trachomatis infection

Genomic DNA was extracted from swabs using the QIAamp DNA mini kit (Qiagen, Manchester, UK). Proprietary lysis buffer and proteinase K were added directly to the specimen collection tube and incubated for 1 hour at 56 °C to lyse the specimen. The DNA was then bound to a silica spin column, washed, and eluted into  $100\mu$ L Tris-Cl EDTA. One 8 $\mu$ L aliquot was tested for conserved regions of *Ct* plasmid open reading frame 2 (diagnostic target) and human 30kDa ribosomal RNA subunit (RPP30; endogenous control target) using an in-house ddPCR, described elsewhere [16,20].

### Serological testing for anti-Pgp3 antibodies

An area of each filter paper calibrated to hold  $10\mu$ L of blood was tested using an ELISA for anti-Pgp3 antibodies [14,21]. Serum was eluted from dried blood spots overnight at 4°C into phosphate-buffered saline (PBS) supplemented with 0.3% volume/volume Tween 20 (PBST) and 5% milk powder, for a final serum dilution of 1:50. Immulon<sup>®</sup> 2HB plates were coated with GST-tagged Pgp3 overnight at 4°C. The next day, plates were washed and then blocked with PBST for 1h at 4–10°C. After removal of blocking buffer, 50µL of the serum elution was added to each well and plates incubated at room temperature for 2h on an orbital shaker. After removing unbound antibody with PBST, 50µL of rabbit anti-human IgG (Southern Biotech, Brimingham, USA) diluted 1:32,000 in PBST was added to each well, and plates were incubated for 1h at room temperature on an orbital shaker. After washing with PBST, plates were incubated in the dark at room temperature for 10 minutes with 50µL per well TMB (KPL, Gaithersburg, USA). The reaction was stopped with  $50\mu$ L 1N H<sub>2</sub>SO<sub>4</sub> and the absorbance read at 450nm on a Spectramax M3 plate reader (Molecular Devices, Wokingham UK). Readings were corrected for background by subtracting the average absorbance of three blank wells containing no serum, using Softmax Pro5 software (Molecular Devices). Serial dilutions of hightitre serum in low-titre serum were run on each plate, and all sample ODs were normalized against that of a middle dilution of the high-titre serum. A finite mixture model [21-23] was used to estimate a negative population in the resulting normalised optical density (NOD) values, and specimens were considered anti-Pgp3 positive if their NOD was more than three standard deviations above the mean of the presumed-negative population (a NOD threshold of 0.245).

### Data recording and analysis

Data entry and real-time server upload was via a bespoke android-based Open Data Kit data capture system, developed as part of the GTMP [19]. Participant data were encrypted and stored in a secure server with only the study investigators having access. TF prevalence data were adjusted for the age of those examined, in five-year age bands. TT prevalence data were adjusted for age and gender of those examined, in five-year age bands. The most recent census data [9] were used as the reference dataset for the purposes of undertaking these adjustments. Results are presented in accordance with STrengthening the Reporting of OBservational studies in Epidemiology (STROBE) guidelines (see <u>S1 Checklist</u>) [24].

### Results

### **Clinical findings**

Fieldwork was conducted in November 2015. This coincided with the tenth anniversary of the arrival of Christianity to Kiritimati; as a consequence of the associated celebrations, a majority of families had decamped to local *maneabas*. We examined 406 of the 412 children aged 1–9 years identified as being resident in selected segments; the other six children were absent at the time of the field teams' visits and could not be located by their parents or guardians. Of those examined, 123 children had TF and 16 had TI; all children with TI also had TF. The age-adjusted prevalence of TF in children was 28% (95% CI 24–35). Of 417 adults aged  $\geq$ 15 years enumerated, 416 were examined, with one person refusing to participate. A total of 8 adults were identified to have TT. The age- and gender-adjusted prevalence of TT in adults was 0.2% (95% CI 0.09–0.33%). One person had bilateral TT.

### Ocular Chlamydia trachomatis infection

Swabs were available from 311/406 (77%) children aged 1–9 years: 118 children with TF and/ or TI, and 193 systematically selected children without either of these signs. Infection was significantly associated with signs of active trachoma (Table 1; Pearson's  $X^2 p < 0.0001$ ). Extrapolating the infection prevalence in swabbed children who had neither TF nor TI (13.5%) to all such children (including the 90 who were not swabbed), we expect that 38 of the 283 children who had neither TF nor TI were infected; extrapolating the infection prevalence in swabbed Table 1. Infection with *Chlamydia trachomatis* in 1–9-year-old children, with and without active trachoma in either eye, Kiritimati Island, Kiribati, November 2015. TF = trachomatous inflammation—follicular; TI = trachomatous inflammation—intense; ddPCR = droplet digital polymerase chain reaction.

|                      |              |              | ddPCR        |             |  |
|----------------------|--------------|--------------|--------------|-------------|--|
|                      |              | Negative (%) | Positive (%) | Total (%)   |  |
| Examination findings | No TF or TI  | 167 (86.5)   | 26 (13.5)    | 193 (62.1)  |  |
|                      | TF and/or TI | 60 (50.8)    | 58 (49.2)    | 118 (37.9)  |  |
|                      | Total        | 227 (73.0)   | 84 (27.0)    | 311 (100.0) |  |

https://doi.org/10.1371/journal.pntd.0005863.t001

children with TF and/or TI (49.2%) to all such children (including the 5 who were not swabbed), we expect that 60 of the children with TF and/or TI were infected. This would give an estimated overall prevalence of infection in examined 1–9-year-olds of 24%.

A median of 17,330 RPP30 copies were found per swab, equivalent to approximately 8,500 host cells per swab. Of PCR-positive swabs (n = 84), the median *Ct* load in children with TF and/or TI in either eye was significantly higher than in those with neither TF nor TI (86,450 vs 6,530 plasmid copies/swab, logistic regression p = 0.0057; Fig 2).

#### Anti-Pgp3 antibodies

Blood spots were analysed from 397 children aged 1–9 years; 8 of the remaining 9 examined children did not assent to fingerprick blood collection, and 1 child was not bled because his parents reported a bleeding diathesis. Seropositivity was significantly associated with active trachoma (TF and/or TI in either eye) ( $X^2 p < 0.00001$ ; Table 2). Reactivity to Pgp3 was higher in children with active trachoma than in those with neither TF nor TI (Fig 3), and significantly higher in those with conjunctival *Ct* infection than in those without (logistic regression p < 0.0001, Fig 4). 90.7% of PCR-positive children were seropositive. The age-specific seroprevalence was low in those aged 1 year (7%), increased rapidly between the ages of 1 and 5 years, and was 76% in those aged 9 years (Fig 5). This increase in seropositivity with age was highly



#### C. trachomatis plasmid copies/swab

Fig 2. Load of *Chlamydia trachomatis* infection in 1–9-year-old children with and without active trachoma in either eye, Kiritimati Island, Kiribati, November 2015. Red lines represent the median value, grey boxes represent inter-quartile ranges and whiskers represent the minima and maxima. TF = trachomatous inflammation—follicular; TI = trachomatous inflammation intense.

https://doi.org/10.1371/journal.pntd.0005863.g002

 Table 2. Reactivity to Pgp3 in 1–9-year-old children with and without active trachoma in either eye, Kiritimati Island, Kiribati, November 2015.

 TF = trachomatous inflammation—follicular; TI = trachomatous inflammation—intense.

|                      |              |              | ELISA        |             |  |
|----------------------|--------------|--------------|--------------|-------------|--|
|                      |              | Negative (%) | Positive (%) | Total (%)   |  |
| Examination findings | No TF or TI  | 164 (59.0)   | 114 (41.0)   | 278 (70.0)  |  |
|                      | TF and/or TI | 23 (19.3)    | 96 (80.7)    | 119 (30.0)  |  |
|                      | Total        | 187 (47.1)   | 210 (52.9)   | 397 (100.0) |  |

https://doi.org/10.1371/journal.pntd.0005863.t002

significant (logistic regression p = 0.0007). There was also a significant increase in NOD with age (logistic regression p < 0.0001).

#### Discussion

We have estimated the prevalence of trachoma and current and cumulative infection with *C. trachomatis* on Kiritimati Island. Our GTMP-based approach to sampling via household selection [19] was challenged by the extensive community bonds of the Kiritimati population, which mean that nuclear families tend not to maintain exclusive residence in a single family dwelling. Prolonged spells of communal living in *maneabas* and individual mobility between households make the question, "Where do you live?" difficult for residents to answer. Rather than undertaking a complete census of the island's population and selecting individuals by simple random sampling, we opted for compact segment sampling, including the geographically adjacent *maneaba* or *maneabas*. This was an imperfect solution, in that it will have somewhat biased survey recruitment by including individuals not normally resident within the land area of the selected segment. We did not keep a record of whether individual participants were examined in their home or in a *maneaba*, entering data on all subjects as though they were residents of the same household. A consequence of this was that we were unable to adjust for clustering in our analyses, which may result in our prevalence estimates being less precise than the confidence intervals that we present here. A further



Normalised Optical Density

Fig 3. Normalized optical density (in an ELISA for anti-Pgp3 antibody) in sera from 1–9-year-old children with and without active trachoma in either eye, Kiritimati Island, Kiribati, November 2015. Red lines represent the median value, grey boxes represent inter-quartile range and whiskers represent the minima and maxima.

https://doi.org/10.1371/journal.pntd.0005863.g003



Fig 4. Normalized optical density (in an ELISA for anti-Pgp3 antibody) in sera from 1–9-year-old children who were positive (red spots in panel (a)) and negative (blue spots in panel (a)) for conjunctival *Chlamydia trachomatis* infection by droplet digital PCR, Kiritimati Island, Kiribati, November 2015. In panel (a), the threshold for anti-Pgp3 seropositivity, derived using a finite mixture model, is indicated (dotted line), and within each one-year age band, spots have been moved horizontally in order to separate data points that would otherwise have been superimposed. In panel (b), red lines represent the median values, grey boxes represent inter-quartile ranges, whiskers represent the minima and maxima.

https://doi.org/10.1371/journal.pntd.0005863.g004





https://doi.org/10.1371/journal.pntd.0005863.g005

limitation was the lack of access to ice to keep samples cold in the field, where temperatures peaked at >30 °C each day. We carried samples in insulated boxes, then refrigerated them at the end of the day's work and until they were transported to Manila, Philippines, for repackaging and shipment to London. DNA degradation occurs more rapidly at higher temperatures, and it is therefore possible that our estimates of *Ct* PCR positivity and DNA loads are underestimates, however our own [15] and others' [25] data suggest this degradation results in (qualitative) diagnostic failure only in very low load samples, which are presumably from individuals least likely to pass infection on to contacts [26]. Loss of antibody following storage for several hours at temperatures >30 °C is minimal [27]. Finally, although we took great care to avoid carry-over contamination of swabbed material from one subject to the next, we cannot say with certainty that this did not occur.

Notwithstanding those limitations, our data suggest that trachoma is a public health problem on Kiritimati Island. The prevalence of TF in 1–9-year-olds observed here (28%) is at a level at which WHO recommends implementation of the A, F and E components of the SAFE strategy for at least three years before re-survey [12]. The Kiritimati TF prevalence that we observed is broadly comparable to that (21%) estimated for South Tarawa and Betio of Kiribati in 2012. Whether these two survey findings, taken together, should be extrapolated as the basis for a decision to deliver population-based interventions, including mass drug administration of azithromycin [28], across the country's 18 other inhabited islands, is a question that will require considerable thought. We submit that such an approach is now justified.

The prevalence of TT that we observed in adults—0.21%, though two significant digits may be more than is justified given that there were only 8 cases—is just over the TT threshold of 0.2% specified by WHO as part of the definition for trachoma's elimination as a public health problem [29]. Ongoing provision of TT surgical services on Kiritimati is likely to be necessary, but because of the relatively small total population, the number of people needing such services is likely to be small. The finding that there is a burden of TT here is again consistent with the situation in South Tarawa and Betio, but contrasts with that in several other Pacific Island countries in which trachoma mapping has been undertaken to date, where TT is rare to absent in adults despite moderate prevalences of TF in children.

We found strong evidence for both current *Ct* infection (demonstrated by PCR-positive conjunctival swabs) and widespread prior exposure to *Ct* infection (demonstrated by antibodies to Pgp3) in children on Kiritimati. Both the presence and load of infection was closely associated with active trachoma in this population, as is typical of trachoma-endemic environments in sub-Saharan Africa [30]. In contrast, in trachoma-endemic communities of the Solomon Islands and Fiji, the prevalence of ocular *Ct* infection has recently been shown to be very low; it has been hypothesized that the latter finding may account, in whole or in part, for the scarcity of trichiasis in those environments [13,15,16].

TI is associated with chronic, repeated conjunctival *Ct* infection and incident TS [31-33], the precursor to TT. Of the 406 children aged 1–9 years that we examined, 16 (4%) had TI, compared to 2 (0.2%) of 1135 children aged 1–9 years examined in the Solomon Islands population in which the prevalence of *Ct* was only 1.3% [16]. The high prevalence of antibody positivity found in the children bled on Kiritimati indicates a high prevalence of previous exposure to *Ct* infection, and the steep increase in seroprevalence we observed between those aged 1 and 5 years suggests intense transmission is occurring in that age group [34,35]. These observations all point to a significant force of infection present in the Kiritimati population, making it straightforward to understand the occurrence of TT in adults, and underscoring the need for implementation of interventions to reduce the prevalence and transmission of infection here. As elimination activities progress, re-measuring the prevalence of conjunctival *Ct* infection and prevalence of anti-Pgp3 antibodies in young children will be instructive, with those

longitudinal data likely to help us to further understand the potential utility of these tools for monitoring programmes [36].

Our finding that a substantial proportion of 1–9-year-olds with neither TF nor TI were *Ct*-positive (13%, Table 1) and anti-Pgp3 antibody-positive (41%, Table 2) is not new [37–40], and has been the source of prolonged discussion over how best to evaluate the potential programmatic use of laboratory tests for *Ct* exposure [41–44]. It is certainly true that the WHO simplified trachoma grading scheme is relatively insensitive, ignoring degrees of conjunctival inflammation less marked than five moderately-sized central follicles or pronounced thickening obscuring at least half of the normal deep tarsal blood vessels [8]. Some children who had neither TF nor TI undoubtedly had lesser degrees of inflammation than this. More sensitive, more detailed trachoma grading systems exist [45], but the simplified system was specifically developed because those systems were felt to be too complex for use by non-specialist personnel working at community level [8], and is the current standard against which any proposed new method for evaluating the success of interventions against trachoma should be assessed.

The differences observed in the epidemiology of trachoma here with that in neighbouring countries could be due to a number of different factors. Kiribati's population is predominantly Micronesian, whereas the island nations to the southwest are dominated by peoples of Melanesian descent. The environment is also different: despite the abundant vegetation, fresh water availability is poor throughout Kiribati because infrastructure is inadequate, the evaporation rate is high, droughts are frequent, and the soil is mostly loose and drains quickly [46]. Such conditions are associated with greater risk of trachoma elsewhere: in a national survey of Nigeria, for example, villages with hotter, drier climates have greater risk of TT than cooler, wetter villages [47]. Because the preponderance of people enrolled in this study were examined in *maneabas* rather than in their own homes, collecting data on household-level water and sanitation risk factors for trachoma was not undertaken [48–50], but doing so in future data collection exercises here, with appropriate adjustments in the unit of data collection (to account for communal living), would be of interest.

Our data indicate a need for interventions against trachoma in this population, and will serve as a baseline for ongoing comparative studies of the epidemiology and pathogenesis of trachoma in the Pacific Islands.

#### Supporting information

**S1** Table. Raw data from this study in accordance with journal data availability guidelines. (CSV)

**S1 Checklist. STROBE checklist.** (DOC)

#### Acknowledgments

We are grateful to Dr Teraira Bangao (Kiritimati Hospital) and his team for their warm welcome, generous hospitality and practical assistance; our field workers Ms Beari Tiribo, Ms Tereitabuki Tarangutu, Mrs Antiera Uriam, and Mrs Tauabo Tatitau; Dr Alfred Tonganibeia (Ministry of Health and Medical Services, Kiribati); Dr André Reiffer and Dr Zeke Nukuro (WHO Country Office, Kiribati); Ms Eva Capa (Fred Hollows Foundation); and the residents of Kiritimati.

The Global Trachoma Mapping Project Investigators are: Agatha Aboe (1,11), Liknaw Adamu (4), Wondu Alemayehu (4,5), Menbere Alemu (4), Neal D. E. Alexander (9), Ana Bakhtiari (2,9), Berhanu Bero (4), Simon J. Brooker (1,6), Simon Bush (7,8), Brian K. Chu

(2,9), Paul Courtright (1,3,4,7,11), Michael Dejene (3), Paul M. Emerson (1,6,7), Rebecca M. Flueckiger (2), Allen Foster (1,7), Solomon Gadisa (4), Katherine Gass (6,9), Teshome Gebre (4), Zelalem Habtamu (4), Danny Haddad (1,6,7,8), Erik Harvey (1,6,10), Dominic Haslam (8), Khumbo Kalua (5), Amir B. Kello (4,5), Jonathan D. King (6,10,11), Richard Le Mesurier (4,7), Susan Lewallen (4,11), Thomas M. Lietman (10), Chad MacArthur (6,11), Colin Macleod (3,9), Silvio P. Mariotti (7,11), Anna Massey (8), Els Mathieu (6,11), Siobhain McCullagh (8), Addis Mekasha (4), Tom Millar (4,8), Caleb Mpyet (3,5), Beatriz Muñoz (6,9), Jeremiah Ngondi (1,3,6,11), Stephanie Ogden (6), Alex Pavluck (2,4,10), Joseph Pearce (10), Serge Resnikoff (1), Virginia Sarah (4), Boubacar Sarr (5), Alemayehu Sisay (4), Jennifer L. Smith (11), Anthony W. Solomon (1,2,3,4,5,6,7,8,9,10,11), Jo Thomson (4); Sheila K. West (1,10,11), Rebecca Willis (2,9). 1. Advisory Committee, 2. Information Technology, Geographical Information Systems, and Data Processing, 3. Epidemiological Support, 4. Ethiopia Pilot Team, 5. Master Grader Trainers, 6. Methodologies Working Group, 7. Prioritisation Working Group, 8. Proposal Development, Finances and Logistics, 9. Statistics and Data Analysis, 10. Tools Working Group, 11. Training Working Group

### **Author Contributions**

- **Conceptualization:** Anaseini Cama, Richard T. Le Mesurier, Rabebe Tekeraoi, Anthony W. Solomon.
- Data curation: Robert M. R. Butcher, Rebecca Willis, Ana Bakhtiari.
- Formal analysis: Robert M. R. Butcher, Rebecca Willis, Ana Bakhtiari, Neal D. E. Alexander, Anthony W. Solomon.

Funding acquisition: Chrissy h. Roberts, Anthony W. Solomon.

- **Investigation:** Anaseini Cama, Andreas Müller, Raebwebwe Taoaba, Robert M. R. Butcher, Iakoba Itibita, Stephanie J. Migchelsen, Tokoriri Kiauea, Harry Pickering, Rabebe Tekeraoi, Anthony W. Solomon.
- Methodology: Robert M. R. Butcher, Diana L. Martin, Anthony W. Solomon.
- Project administration: Anaseini Cama, Rabebe Tekeraoi.
- Supervision: Anaseini Cama, Chrissy h. Roberts, Richard T. Le Mesurier, Rabebe Tekeraoi, Anthony W. Solomon.
- Visualization: Robert M. R. Butcher.
- Writing original draft: Anaseini Cama, Andreas Müller, Robert M. R. Butcher, Anthony W. Solomon.
- Writing review & editing: Neal D. E. Alexander, Diana L. Martin, Anthony W. Solomon.

#### References

- 1. Bourne RA, Stevens GA, White RA, Smith JL, Flaxman SR, Price H, et al. Causes of vision loss worldwide, 1990–2010: a systematic analysis. Lancet Glob Heal. 2013; 1: e339–49.
- 2. World Health Assembly. Resolution WHA51.11. Global elimination of blinding trachoma. Geneva, Switzerland: 51st World Health Assembly, 16 May; 1998.
- 3. Taylor H. Trachoma in Australia. Med J Aust. 2001; 175: 371–2. PMID: 11700815
- 4. Ward B. The prevalence of active trachoma in Fiji. Am J Ophthalmol. 1965; 59: 458-63.
- 5. Macallan AF. Trachoma in the British Colonial Empire—its relation to blindness, the existing means of relief, means of prophylaxis. Br J Ophthalmol. 1934; 18: 625–45. PMID: <u>18169234</u>

- 6. Mann I. Some Islands. Climate, race, culture and eye disease. 1st ed. Springfield, IL, USA: Charles C Thomas; 1966.
- Mathew AA, Keeffe JE, Le Mesurier RT, Taylor HR. Trachoma in the Pacific Islands: evidence from Trachoma Rapid Assessment. Br J Ophthalmol. 2009; 93: 866–70. https://doi.org/10.1136/bjo.2008. 151720 PMID: 19174394
- Thylefors B, Dawson CR, Jones BR, West SK, Taylor HR. A simple system for the assessment of trachoma and its complications. Bull World Health Organ. 1987; 65: 477–83. PMID: 3500800
- 9. Kiribati National Statistics Office. Report on the Kiribati 2010 Cenus of Population and Housing. Bairiki: Ministry of Finance and Economic Planning, Kiribati; 2012.
- International Agency for the Prevention of Blindness. Trachoma Mapping in the Pacific. Melbourne, Australia; 2013.
- Francis V, Turner V. Acheiving community support for trachoma control: a guide for district health work. Geneva, Switzerland: World Health Organization; 1995.
- Solomon A, Zondervan M, Kuper H, Buchan J, Mabey D, Foster A. Trachoma control: a guide for program managers. Geneva, Switzerland: World Health Organization; 2006.
- Taleo F, Macleod CK, Marks M, Sokana O, Last AR, Willis R, et al. Integrated mapping of yaws and trachoma in the five northern-most provinces Vanuatu. PLoS Negl Trop Dis. 2017; 11: e0005267. <u>https:// doi.org/10.1371/journal.pntd.0005267 PMID: 28118354</u>
- Cocks N, Rainima-Qaniuci M, Yalen C, Macleod C, Nakolinivalu A, Migchelsen S, et al. Community seroprevalence survey for yaws and trachoma in the Western Division of Fiji. Trans R Soc Trop Med Hyg. 2016; 110: 582–587. https://doi.org/10.1093/trstmh/trw069 PMID: 27852877
- Macleod CK, Butcher R, Mudaliar UU, Natutusau K, Pavluck AL, Willis R, et al. Low prevalence of ocular Chlamydia trachomatis infection and active trachoma in the Western Division of Fiji. PLoS Negl Trop Dis. 2016; 10: e0004798. https://doi.org/10.1371/journal.pntd.0004798 PMID: 27404379
- Butcher RMR, Sokana O, Jack K, Macleod CK, Marks ME, Kalae E, et al. Low Prevalence of Conjunctival Infection with Chlamydia trachomatis in a Treatment-Naïve Trachoma-Endemic Region of the Solomon Islands. PLoS Negl Trop Dis. 2016; 10: e0004863. <u>https://doi.org/10.1371/journal.pntd.0004863</u> PMID: 27603015
- 17. Sokana O, Macleod CK, Jack K, Butcher R, Marks M, Willis R, et al. Mapping trachoma in The Solomon Islands—results from the Global Trachoma Mapping Project. Ophthalmic Epidemiol. 2016; 23: 15–21. https://doi.org/10.1080/09286586.2016.1238946 PMID: 27937043
- Kirkwood B, Sterne JA. Calculation of required sample size. Essentail Medical Statistics. 2nd ed. Oxford, UK: Blackwell Publishing Ltd; 2003. pp. 413–428.
- Solomon AW, Pavluck AL, Courtright P, Aboe A, Adamu L, Alemayehu W, et al. The Global Trachoma Mapping Project: Methodology of a 34-Country Population-Based Study. Ophthalmic Epidemiol. 2015; 22: 214–25. https://doi.org/10.3109/09286586.2015.1037401 PMID: 26158580
- Roberts Ch, Last A, Molina-Gonzalez S, Cassama E, Butcher R, Nabicassa M, et al. Development and Evaluation of a Next-Generation Digital PCR Diagnostic Assay for Ocular Chlamydia trachomatis Infections. J Clin Microbiol. 2013; 51: 2195–203. https://doi.org/10.1128/JCM.00622-13 PMID: 23637300
- Migchelsen SJ, Martin DL, Southisombath K, Turyaguma P, Heggen A, Rubangakene PP, et al. Defining Seropositivity Thresholds for Use in Trachoma Elimination Studies. PLoS Negl Trop Dis. 2017; 11: e0005230. https://doi.org/10.1371/journal.pntd.0005230 PMID: 28099433
- 22. McLachlan G, Peel D. Finite Mixture Models. 1st ed. Hoboken, USA: John Wiley & Sons, Ltd; 2000.
- 23. Benaglia T, Chauveau D, Hunter DR, Young DS. mixtools: An R package for analyzing mixture models. J Stat Softw. 2009; 32: 1–29.
- von Elm E, Altman DG, Egger M, Pocock SJ, Gøtzsche PC, Vandenbroucke JP. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. Lancet. 2007; 370: 1453–7. <u>https://doi.org/10.1016/S0140-6736(07)61602-X</u> PMID: 18064739
- 25. Dize L, West SK, Mkocha H, Quinn TC, Gaydos CA. Evaluation of pooled ocular and vaginal swabs by the Cepheid GeneXpert CT/NG assay for the detection of Chlamydia trachomatis and Neisseria gonorrhoeae compared to the GenProbe Aptima Combo 2 Assay. Diagn Microbiol Infect Dis. 2015; 81: 102– 104. https://doi.org/10.1016/j.diagmicrobio.2014.11.010 PMID: 25497459
- Last AR, Burr SE, Alexander N, Harding-Esch EM, Roberts Ch., Nabicassa M, et al. Spatial clustering of high load ocular Chlamydia trachomatis infection in trachoma: A cross-sectional population-based study. Pathog Dis. 2017; ftx050. https://doi.org/10.1093/femspd/ftx050 PMID: 28472466
- Corran PH, Cook J, Lynch C, Leendertse H, Manjurano A, Griffin J, et al. Dried blood spots as a source of anti-malarial antibodies for epidemiological studies. Malar J. 2008; 7: 195. https://doi.org/10.1186/ 1475-2875-7-195 PMID: 18826573

- Evans JR, Solomon AW. Antibiotics for trachoma. Cochrane Database Syst Rev. 2011; CD001860. https://doi.org/10.1002/14651858.CD001860.pub3 PMID: 21412875
- **29.** World Health Organization. Validation of elimination of trachoma as a public health problem (WHO/ HTM/NTD/2016.8). Geneva, Switzerland; 2016.
- Solomon AW, Holland MJ, Burton MJ, West SK, Alexander NDE, Aguirre A, et al. Strategies for control of trachoma: observational study with quantitative PCR. Lancet. 2003; 362: 198–204. <u>https://doi.org/10. 1016/S0140-6736(03)13909-8 PMID: 12885481</u>
- Wolle MA, Muñoz BE, Mkocha H, West SK. Constant ocular infection with Chlamydia trachomatis predicts risk of scarring in children in Tanzania. Ophthalmology. 2009; 116: 243–7. https://doi.org/10.1016/ j.ophtha.2008.09.011 PMID: 19091415
- Wolle MA, Muñoz B, Mkocha H, West SK. Age, sex, and cohort effects in a longitudinal study of trachomatous scarring. Invest Ophthalmol Vis Sci. 2009; 50: 592–6. https://doi.org/10.1167/iovs.08-2414 PMID: 18936137
- Burton MJ, Rajak SN, Hu VH, Ramadhani A, Habtamu E, Massae P, et al. Pathogenesis of progressive scarring trachoma in ethiopia and Tanzania and its implications for disease control: two cohort studies. PLoS Negl Trop Dis. 2015; 9: e0003763. <u>https://doi.org/10.1371/journal.pntd.0003763</u> PMID: 25970613
- Martin DL, Wiegand R, Goodhew B, Lammie P, Black CM, West S, et al. Serological Measures of Trachoma Transmission Intensity. Sci Rep. 2015; 5: 18532. https://doi.org/10.1038/srep18532 PMID: 26687891
- Martin DL, Bid R, Sandi F, Goodhew EB, Massae PA, Lasway A, et al. Serology for Trachoma Surveillance after Cessation of Mass Drug Administration. PLoS Negl Trop Dis. 2015; 9: e0003555. <u>https://doi.org/10.1371/journal.pntd.0003555 PMID: 25714363</u>
- West SK, Munoz B, Weaver J, Mrango Z, Dize L, Gaydos C, et al. Can We Use Antibodies to Chlamydia trachomatis as a Surveillance Tool for National Trachoma Control Programs? Results from a District Survey. PLoS Negl Trop Dis. 2016; 10: e0004352. <u>https://doi.org/10.1371/journal.pntd.0004352</u> PMID: 26771906
- Solomon AW, Peeling RW, Foster A, Mabey DCW. Diagnosis and assessment of trachoma. Clin Microbiol Rev. 2004; 17: 982–1011. https://doi.org/10.1128/CMR.17.4.982-1011.2004 PMID: 15489358
- Ramadhani AM, Derrick T, Macleod D, Holland MJ, Burton MJ. The Relationship between Active Trachoma and Ocular Chlamydia trachomatis Infection before and after Mass Antibiotic Treatment. PLoS Negl Trop Dis. 2016; 10. https://doi.org/10.1371/journal.pntd.0005080 PMID: 27783678
- Goodhew EB, Priest JW, Moss DM, Zhong G, Munoz B, Mkocha H, et al. CT694 and pgp3 as Serological Tools for Monitoring Trachoma Programs. PLoS Negl Trop Dis. 2012; 6. <u>https://doi.org/10.1371/</u> journal.pntd.0001873 PMID: 23133684
- Goodhew EB, Morgan SM, Switzer AJ, Munoz B, Dize L, Gaydos C, et al. Longitudinal analysis of antibody responses to trachoma antigens before and after mass drug administration. BMC Infect Dis. 2014; 14: 216. https://doi.org/10.1186/1471-2334-14-216 PMID: 24755001
- **41.** Wright HR, Taylor HR. Clinical examination and laboratory tests for estimation of trachoma prevalence in a remote setting: what are they really telling us? Lancet Infect Dis. 2005; 5: 313–20. <u>https://doi.org/10.1016/S1473-3099(05)70116-X PMID: 15854887</u>
- Solomon AW, Foster A, Mabey DCW. Clinical examination versus Chlamydia trachomatis assays to guide antibiotic use in trachoma control programmes. Lancet Infect Dis. 2006; 6: 5–6. https://doi.org/10. 1016/S1473-3099(05)70304-2 PMID: 16377526
- **43.** Michel C-E, Roper KG, Divena MA, Lee HH, Taylor HR. Correlation of clinical trachoma and infection in Aboriginal communities. PLoS Neglected Trop Dis. 2011; 5: e986. <u>https://doi.org/10.1371/journal.pntd.</u> 0000986 PMID: 21423648
- Michel C-E, Solomon AW, Magbanua JP V, Massae PA, Huang L, Mosha J, et al. Field evaluation of a rapid point-of-care assay for targeting antibiotic treatment for trachoma control: a comparative study. Lancet. 2006; 367: 1585–1590. https://doi.org/10.1016/S0140-6736(06)68695-9 PMID: 16698411
- Dawson CR, Jones BR, Tarizzo ML, World Health Organization. Guide to trachoma control in programmes for the prevention of blindness. Geneva, Switzerland: World Health Organization; 1981.
- **46.** Thomas FR. Kiribati: some aspects of human ecology, forty years later. National Museum of Natural History, Smithsonian Institution. Washington, D.C.; 2003.
- 47. Smith JL, Sivasubramaniam S, Rabiu MM, Kyari F, Solomon AW, Gilbert C. Multilevel Analysis of Trachomatous Trichiasis and Corneal Opacity in Nigeria: The Role of Environmental and Climatic Risk Factors on the Distribution of Disease. PLoS Negl Trop Dis. 2015; 9: e0003826. <u>https://doi.org/10.1371/journal.pntd.0003826</u> PMID: 26222549

- Bero B, Macleod CK, Alemayehu W, Gadisa S, Abajobir A, Adamu Y, et al. Prevalence of and Risk Factors for Trachoma in Oromia Regional State of Ethiopia: Results of 79 Population-Based Prevalence Surveys Conducted with the Global Trachoma Mapping Project. Ophthalmic Epidemiol. 2016; 23: 392– 405. https://doi.org/10.1080/09286586.2016.1243717 PMID: 27820657
- 49. Elshafie BE, Osman KH, Macleod C, Hassan A, Bush S, Dejene M, et al. The Epidemiology of Trachoma in Darfur States and Khartoum State, Sudan: Results of 32 Population-Based Prevalence Surveys. Ophthalmic Epidemiol. 2016; 23: 381–391. <u>https://doi.org/10.1080/09286586.2016.1243718</u> PMID: 27841721
- 50. Adera TH, Macleod C, Endriyas M, Dejene M, Willis R, Chu BK, et al. Prevalence of and Risk Factors for Trachoma in Southern Nations, Nationalities, and Peoples' Region, Ethiopia: Results of 40 Population-Based Prevalence Surveys Carried Out with the Global Trachoma Mapping Project. Ophthalmic Epidemiol. 2016; 23: 1–7.